 
CENWP-OP-B                                                                                                                                   25 June 2001

MEMORANDUM FOR CENWP-OP, ATTN: Gary Johnson

SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for week 25 of 2001, which covers the period 17 through 23 June.
1.  OPERATION SUMMARY.

1.1 Daily river flow averaged 103.1 to 151.0kcfs.  Daily spill averaged 0.0 kcfs.  Daily project tailwater elevation averaged 10.0’ to 13.2’ msl.  Daily project forebay elevation averaged 73.3’ to 74.4’ msl.  River temperature averaged 62 to 64° F.  

1.2 Spill bays 1 and 18, which supply attraction flow for Cascades Island and Bradford Island ladders respectively, remain open 4” during daylight hours.    

1.3 Unit 3 remained out of service for turbine rehab. Unit 17 remains out of service for its four-year overhaul. Unit 15 FGE testing resumed 15 June.  The unit will return to standby each night following FGE testing.  Unit 5 went out of service for rehab work at 0754 on 18 June.  Unit 2 returned to service at 2212 on 22 June.  Unit 15 went out of service at 0825 on 23 June after the STS in the B slot became impossible to pick with the crane.  FGE testing is suspended until early next week when divers arrive to help retrieve the screen.  

1.4 Powerhouse Two fish units remain in service.  They provide attraction flow for the Washington Shore fish ladder.  The units are placed on standby from approximately 0000-0300 nightly if drawdown exceeds 1.0’.  Both Fish Units were out of service from 0842 until 0918 on 20 June to facilitate video inspection of the triangle area.

2.  ACTIVITIES. 
2.1 AWS Closures.   FV4-3 and FV4-4 at B-branch and FV5-3 and FV5-4 are all operating in manual until the new computer system is installed.  FV3-9 and BI exit were closed for cleaning from 1905 until 1925 on 19 June.  

2.2 STS/VBS Inspections.  Two units were video inspected this reporting week.

Unit
Date
Run Hours
Comments

09
6/20
547
Good

10
6/20
741
Good

Table 1.  STS/VBS Inspection Data.


2.3 Dewatering and Fish Salvage.  On 20 June, Project Biologists removed six lamprey, six live sturgeon, and three dead sturgeon from the draft tube of Unit 5.  All fish were released at the Hamilton Island boat launch.

3.  RESEARCH.

3.1 WDFW Fish Counting.  WDFW continues 16-hour visual counts and 8-hour video counts at Bradford Island and Washington Shore ladders.

3.2 NMFS Saltwater rearing. The Saltwater rearing facility is in operation.  A seawater spill occurred on graveyard shift 18 June.  See Miscellaneous 5.7.
3.3 OCFWRU Barging condition study.  Began 24 April.

3.4 PSMFC Adult PIT.  Looking at A and B branch and Cascade Island for noise interference.  Analyzing data collected at the Washington Shore ladder.

3.5 PSMFC Smolt Monitoring Program.  Collection continues 24h/d at the monitoring facility.  24 h/d flat plate monitoring at DSM1.

3.6 WES Hydroacoustic Monitoring.  Monitoring began 20 April.

3.7 UI Barging condition study. They began sampling juvenile fish from transport barges on 22 April.

3.8 USGS Juvenile Radiotracking.  Radio tagging began 1 April.  Tracking in progress as well.

3.9 USGS Adult EMG tracking.  Finished releasing fish 10 June.  Will continue tracking until 30 June.

3.10 UI Adult Fallback. UI Adult Radiotracking/Tagging.  PCC entrance gates at Powerhouse One are being operated according to UI request until 31 October.  Tagging of Chinook began at the AFC&MF 2 April and will continue seven days a week until the end of September. 

3.11 CRITFC/ODFW.  Tagging Spring Chinook and steelhead on Tuesdays and Thursdays.

3.12 WDFW.  Tagging Spring Chinook with jaw tags in conjunction with their tooth-net studies.

3.13 NMFS FGE/OPE.  Work began 23 April.  Spring testing ended ~ 23 May.  Summer tests began 11 June.  

3.14 NMFS Lamprey.  Collection of Lamprey began June 1.  Spillway ladder entrances manually operated at night for a .5’ differential.  NMFS is holding extra lamprey for the Warm Springs Tribe, and collecting fish for USGS.

3.15 Pelfrey Sturgeon Hatchery.  Collecting broodstock in PH1 tailrace. 
4.  FISHWAYS. 
4.1 Project biologists inspected every day (17-23 June) this reporting week.  NMFS inspected the fishways on 21 June.

4.2 Fish Passage Plan Observations.

DATE
LOCATion
FPP Deviation
CAUSE
RESPONSE

All week
PH1and PH2 collection channel electronic velocity meters.
PH1 is reading a low velocity, PH2 is not yet working.
Unknown.
New velocity meters installed in channel, need more cable at PH2, PH1 may need calibrated.

6/20-6/23
FG3-28
Out of service
Manually closed because of displaced diffuser valve cover.   See 5.1 below.
New covers are being designed, and will be installed during winter maintenance.

6/20
FG3-25
Not in correct position for TW elevation.
Computer system not yet installed, all diffusers at B-Branch are operated in manual
Operator corrected diffuser positioning.

6/20
FG6-12
Not in correct position for TW elevation.
Computer system not yet installed, all diffusers at B-Branch are operated in manual
Operator corrected diffuser positioning

6/20
DSM2 channel
Elevation low.
Unknown
No action taken.

6/19
11C-S, F1B-N
Closed though PLC read opened.
Unknown, possibly faulty proximity switches.
Operators opened orifices in local.  Maintenance will look at them.

6/18-6/19
WG1, WG65
Not in correct position for TW elevation.
Unknown
Operators manually closed the gates.

4.3 Fish Ladder Temperatures.  Temperatures are measured 3 weirs down from the count station at Bradford Island and at the junction of the UMT and the main ladder at Washington shore.

Date
Wash. Shore
Bradford Is.

6/17
16.4
16.3

6/18
16.4
16.3

6/19
16.6
16.5

6/20
17.0
16.9

6/21
17.3
17.2

6/22
17.5
17.4

6/23
17.5
17.4

4.4 Sensor Calibration.  Water level sensors were checked for calibration on 21 June this reporting week.  

4.5 Zebra Mussel Monitoring.  The zebra mussel monitoring station was inspected and showed no signs of colonization this reporting week.

5. MISCELLANEOUS

5.1 Mid-season collection channel dives revealed that diffuser valve cover FG3-28 in Bradford Island B-branch has come away from the wall, exposing a 13” gap between the floor grating and the wall.  This has been seen many times in the past, as the spalled concrete is not capable of anchoring the valve covers anymore.  The project is designing a new type of cover that will not rely on the wall for an anchor point.

5.2 Injured and dead juvenile fish seen at the JMF decreased this week according to PSMFC.  Project biologists released a water filled beach ball into the transport pipe at 1015 on 20 June.  The ball exited the pipe at the JMF 13 hours later.  PSMFC released air filled and water filled latex gloves into the pipe.  The water filled gloves exited 40 minutes later.  The air filled gloves still have not been recovered.

5.3 The JMF was put into bypass mode from 0733 until 1600 on 20 June to facilitate PDS weir modification work and the general holding tank crowder work.  PSMFC resumed sampling by 1200 on 21 June.

5.4 FV3-9 was malfunctioning and so was put into manual at 2312 on 19 June.  Electricians worked on the problem and returned the valve to auto at 1500 on 20 June.

5.5 New velocity meters were installed during the mid-season dives 13 and 14 June.  PH1 meter is operating. PH2 meter is not yet operating because of a problem with the data transmission cable.

5.6 Spill stopped at 0000 16 June.  All bays are on seal during evening hours.  Bays 1 and 18 will remain open 4” during daylight hours.

5.7 NMFS Saltwater Rearing project spilled 800-1000 gallons of seawater during the graveyard shift on 18 June.  The seawater overflowed the holding tank for processed (filtered) water and spilled into the compacted gravel fill beneath the Sprung Structure. 

                                                                                                                DEBORAH L. CHENOWETH

                                 

                                                             Operations Project Manager
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